Author’s response to reviews

Title: Low vitamin D levels are associated with cognitive impairment in patients with hashimoto thyroiditis

Authors:

Jun Xu (jiaxingxujun@163.com)
Xiang-yun Zhu (850321zxy@163.com)
Hui Sun (361809380@qq.com)
Xiao-qin Xu (175563760@qq.com)
Song-ao Xu (651271731@qq.com)
Yuan Suo (364824199@qq.com)
Li-jun Cao (867262545@qq.com)
Qiang Zhou (zhouqiang8005@163.com)
Hui-jie Yu (yuhuijie88@sina.cn)
Wei-zhong Cao (caowz573@sina.com)

Version: 1 Date: 15 Aug 2018

Author’s response to reviews:

Dear Editor,

Thank you very much for your letter regarding our manuscript entitled “Low vitamin D levels are associated with cognitive impairment in patients with hashimoto thyroiditis”. We are truly grateful to the technical comments from you. Those comments are all valuable and very helpful for revising and improving our paper.

Accordingly, we have carefully studied comments and revised our manuscript accordingly. In addition, we have carefully read the editorial policies of respected BMC Endocrine Disorders and revised our manuscript as necessary.
We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

Yours sincerely,

Jun Xu

Corresponding author